Author's response to reviews

Title: Profile of blood cell abnormalities of treatment-naive HIV patients in the Yaounde University Teaching Hospital, Cameroon.

Authors:

Paul N Wankah (wankahpaul@gmail.com)
Claude T Tayou (claudetayou@yahoo.fr)
Dora NS Mbanya (dmbanya1@yahoo.co.uk)

Version: 2
Date: 15 March 2014

Author's response to reviews:

Cover letter of revision
1) The institutions, countries and e-mail addresses were updated for all authors.
2) Line numbering was added.
3) The competing interest section was added.
4) The authors' contribution was written in the format suggested by the journal.